Author's response to reviews

Title: Ethical challenges related to elder care. High level decision-makers' experiences

Authors:

Anna-Greta Mamhidir (anna-greta.mamhidir@lg.se)
Mona Kihlgren (mona.kihlgren@orebroll.se)
Venke Sorlie (venke.soerlie@hibo.no)

Version: 7 Date: 2 April 2007

Author's response to reviews:

2007-04-02

Cover letter

Dear editorial production team at BMC Medical Ethics,

I am sorry that we have not written clearly enough to fulfill the formatting changes and we hope that these clarifications will be acceptable.

The following formatting changes are made according to your requests for MS 1370148345109003.

1. We have now changed the text at page 24, below the heading - Authors' contributing. The text is now as follows:
AGM participated in the design of the study, carried out the interviews, participated in the analysis, completed and approved to the final manuscript.
MK and VS participated in the design of the study, read the interviews, participated in the analysis, helped to complete and approved the final manuscript.

2. The heading Discussion and Conclusions is now changed to Discussion and conclusions.

Best regards
Anna-Greta Mamhidir and co-authors.